Author's response to reviews

**Title:** Is there a role for consolidative radiotherapy in the treatment of aggressive and localized Non-Hodgkin Lymphoma? A systematic review with meta-analysis

**Authors:**

Lucas V dos Santos (lucassvsantos@yahoo.com)
João Paulo SN Lima (jpsnlima@yahoo.com.br)
Carmen SP Lima (carmenl@fcm.unicamp.br)
Emma C Sasse (emma.sasse@uol.com.br)
Andre D Sasse (sasse@cevon.com.br)

**Version:** 2 **Date:** 8 November 2011

**Author's response to reviews:**

Dear Editor,

We are sending you the revised manuscript entitled "Is there a role for consolidative radiotherapy in the treatment of aggressive and localized Non-Hodgkin Lymphoma? A systematic review with meta-analysis". It was prepared according to the PRISMA statement.

Sincerely,